Author's response to reviews

Title: Oral Status, Oral Health knowledge, Attitudes and Behavior in rural children of Shanxi, western China

Authors:

Gao Jianghong (mamaluck@sohu.com)
Ruan Jianping (851432881@qq.com)
Zhao Lin (851432881@qq.com)
Zhou Hong (851432881@qq.com)
Huang Ruizhe (851432881@qq.com)
Tian Jiangang (851432881@qq.com)

Version: 2
Date: 3 June 2014

Author's response to reviews: see over
Dear editor

Dear editor:

Thanks a lot for having reviewed our manuscript. Now we have revised the manuscript according to your comments. Most of the revisions are in the manuscript. We would like to resubmit the manuscript entitled “Oral Status, Oral Health knowledge, Attitudes and Behavior in rural children of Shanxi, western China” in the journal of BMC Oral Health.

1. Title Page. It had contained the names, institutions, countries and email addresses of all authors, and the full postal address of the submitting author.

2. Line Numbering. We had revised our manuscript to include line and page numbers.

3. Ethics. A statement detailing what ethical approval has been obtained had been added into the manuscripts. The study was consent by participants’ parent and a statement about this had been added into the manuscript.

4. Competing Interests. ‘Competing interests’ section had been placed after the Conclusions. The authors declare that we have no competing interests.

5. Authors’ Contributions. Authors’ Contributions section had been placed after the Competing Interests section.

Thank you very much once again for your comments.

Sincerely yours,

Jianghong Gao